Author's response to reviews

Title: Antinociceptive and anti-inflammatory activities of Geranium bellum and its isolated compounds

Authors:

Claudia Gonzalez (cvg09@yahoo.com)
Raquel Cariño (raquelcarchortes@gmail.com)
Juan A Gayosso (gayitojuan@yahoo.com)
Mario I Ortiz (mario_i_ortiz@hotmail.com)
Minarda De la O (minidelao@hotmail.com)
Diana A Altamirano (baezaltaandrea@hotmail.com)
Luis Jimenez (luis.jimenez@ieee.org)
Mirandeli Bautista (mirandeli@hotmail.com)

Version: 2 Date: 29 July 2013

Author's response to reviews: see over
I am submitting the manuscript entitled “Antinociceptive and anti-inflammatory activities of *Geranium bellum* and its isolated compounds”. This work provides evidences that extract from *Geranium bellum* Rose aerial parts acts as an effective analgesic in traditional medicine, and to establish for the first time the local and systemic anti-inflammatory and antinociceptive effects of the extract and compounds of *Geranium bellum*, in vivo.

In this new version, as you ask, authors' Contributions and ethics statement was include. We also complete the background section.

We very much hope that you will find it suitable for publication in the Complementary & Alternative Medicine journal.

Sincerely,

PhD. Mirandeli Bautista Ávila

Área Académica de Medicina del Instituto de Ciencias de la Salud Universidad Autónoma del Estado de Hidalgo. Ex Hacienda la Concepción Carr. Pachuca-Actopan s/n Tilcuautla, Hidalgo. 42160, México.

Phone Number: +52-77-1717-2000 Ext. 5106 Fax Number: +52-77-1717-2000 Ext.

5113 E-mail address: mirandeli@hotmail.com (M. Bautista)